Involving the patient's perspective and preferences concerning coronary angiography and percutaneous coronary intervention



Marlies M. Kok¹, MD, PhD; Clemens von Birgelen^{1,2*}, MD, PhD

1. Department of Cardiology, Thoraxcentrum Twente, Medisch Spectrum Twente, Enschede, the Netherlands; 2. Department of Health Technology and Services Research, Technical Medical Centre, University of Twente, Enschede, the Netherlands

The interventional cardiology community is currently increasing its efforts to involve the patient's perspective, which includes the process of the informed consent and the medical treatment itself, the implementation of novel therapies, and the assessment of the consequences of treatment¹. Moreover, recent international guidelines recommend sharing decision making by involving the patient's preferences if alternative therapies are similarly suitable^{2,3}. Shared decision making is a process in which a physician offers information to the patient, elicits the patient's preferences, and then comes to a decision in true partnership and congruence with the patient⁴. This implies that suitable information on indication, risks, and long-term consequences of various treatment options have been provided. Sharing decision making between patient and physicians is integral to contemporary medical practice⁵. The majority of patients do welcome being (partly) in control of the next diagnostic and therapeutic steps that will be taken, while for some patients the process may be emotionally demanding, laborious, or not appreciated⁵⁻⁷. Nevertheless, in the field

of interventional cardiology, many clinical decisions could – or should – involve elements of shared decision making.

The choice of vascular access for coronary angiography or percutaneous coronary intervention (PCI) is well suited to shared decision making, which is appreciated by most patients⁷. Previous research has shown that vascular access via the radial artery reduced the risk of major bleedings and all-cause mortality as compared to transfemoral access. Therefore, current guidelines recommend using the radial access as the standard approach for all PCI procedures, unless there are overriding procedural considerations³. Various aspects may be relevant for discussing the choice of vascular access with the patient, such as the individual bleeding risk, the presence of obstructive peripheral arterial disease, the history of coronary artery bypass grafting (CABG) or peripheral vascular interventions, the accessibility of certain puncture sites, the operator's personal experience, the patient's capability to lie flat on his/her back for hours, the patient's experience with previous coronary procedures (e.g., including pain, spasm, and

*Corresponding author: Department of Cardiology, Thoraxcentrum Twente, Medisch Spectrum Twente, Postbus 50.000, 7500 KA Enschede, the Netherlands. E-mail: c.vonbirgelen@mst.nl

© Europa Digital & Publishing 2020. All rights reserved.

DOI: 10.4244/EIJV15I14A221

mobilisation) and, last but not least, the patient's job, hobbies, and handedness. In a recent patient preference study⁷, patients appreciated lower bleeding risk and early ambulation, inherent to radial access. Moreover, patients who had previous experience with both vascular access routes generally preferred the radial access. In addition, previous experience with a single vascular access route had a major impact on patients' preference⁷, which should be considered when discussing this issue with patients.

The discussion on whether to proceed with PCI for a stable coronary syndrome is particularly well suited to shared decision making⁴. Current guidelines not only suggest involving the patient's preferences for anti-ischaemic pharmacological treatment² but also recommend that Heart Teams develop individualised treatment concepts with respect for the well-informed patients' preferences in the presence of stable angina and obstructive coronary artery disease³. Nevertheless, in communications between cardiologists and patients about coronary angiography and PCI, there is a risk that explicit or implicit overstatement of benefits and understatement of risks may contribute to a misperception of potential benefits8. Moreover, individual communication styles may hinder patient understanding and participation in decision making⁸, and informed decision making is often incomplete9. Although cardiologists generally have a realistic view about the benefits expected from PCI, many patients with stable angina erroneously believe that PCI will prevent future (fatal) myocardial infarctions¹⁰. Moreover, after a standard informed consent the patients' retention of information is low¹¹.

Audiovisual tools and decision aids may help to avoid many of these problems, and they can structure conversations between patients and physicians during outpatient consultations¹². The use of such tools allows standardising the informed consent process and is likely to suit many patients; however, some patients might require tailor-made conversations, such as those with language barriers, problems understanding the content being presented, hearing difficulties, an increased timidity, or a compelling need to obtain more extensive information. In the PREPARED study, an internet-based decision aid was shown to be associated with improved patient knowledge and greater interest in shared decision making but did not lead to an overall change in patient preferences¹³. This cluster-randomised study demonstrated the feasibility of integrating patient preference information into clinical care; however, providing preferences to clinicians did not improve concordance between patients' preference and actual treatment¹³. PCI Choice, another randomised study, revealed that a decision aid that was designed to discuss the choice between treatment with PCI versus optimal medical therapy increased the patients' knowledge but had no impact on decisional quality and the patients' engagement in shared decision making¹⁴. This could be partly related to both the physicians' reluctance to modify established practice patterns and their misconception that the decision aid serves primarily as a patient education tool¹⁵.

Another field of decision making that should involve patients' perspectives is when Heart Teams discuss revascularisation

strategies in patients with severe obstructive multivessel coronary artery disease3. During recent years, the differences in clinical outcomes between PCI and CABG have been reduced³. Moreover, to some patients the avoidance of a phase of major morbidity and the preservation or recovery of quality of life may be of greater importance than the life-prolonging effect of the treatment. This may apply particularly to patients who fear complications due to severe comorbidities, frailty, or advanced age, as well as to patients whose life expectancy is relatively low and unlikely to be markedly changed by the revascularisation procedure. Some patients may refuse to undergo CABG for other reasons, as has been shown in a prospective study that evaluated patient-centred decision making on revascularisation strategy in patients with multivessel or unprotected left main coronary artery disease¹⁶. Although this study used a standardised protocol that favoured CABG, only 38% of all patients were consented for CABG, and the majority of the patients preferred PCI. Reasons for declining CABG included a straightforward refusal of open heart surgery (68%), mild presentation of anginal symptoms (28%), and low self-confidence to expect long-term survival (26%)¹⁶.

Several studies have shown that many patients are not worried about a somewhat increased risk of repeat PCI17. On the other hand, the significantly longer hospitalisation and recovery period after CABG and the patients' morbidity during this stage may be the most important reasons why a proportion of all candidates for CABG refuses the surgical revascularisation. This was suggested by a study in patients with a history of both CABG and at least two PCI procedures (10% even had two CABG procedures), who were faced with a hypothetical coronary revascularisation decision¹⁸. The majority of these "expert patients" preferred staged PCI over CABG, although the period without anginal symptoms had been significantly longer after CABG than after PCI¹⁸. To examine the perspectives on adverse cardiovascular events of patients who undergo PCI or CABG, the PREVAS study used a stated preference elicitation method (i.e., best-worst scaling) and found that the patients considered a hypothetical need for CABG during one-year follow-up much more serious than a repeat PCI¹⁷. In the OPINION study, patients with stable coronary disease completed a case vignette questionnaire on a hypothetical significant left main stenosis amenable to PCI or CABG. When given the choice, most (89%) study participants preferred PCI over CABG¹⁹. Considering these findings, it is fair to state that a substantial proportion of suitable candidates for CABG may (tend to) refuse surgery, primarily as these patients believe that this type of surgical intervention and the recovery thereafter might put too heavy a strain on them.

Results from a survey among patients undergoing non-emergent coronary angiography with possible PCI showed that many patients consider avoiding the drawbacks of dual antiplatelet therapy (DAPT) as important as avoiding a repeat PCI²⁰. Nevertheless, the use of an individualised shared decision-making tool for stent selection was shown to have no impact on stent selection or the concordance between patients' stent preference and the stent received²¹. This led to the conclusion that physician-level barriers to shared decision making may exist²¹. While this might be true, one cannot exclude that the interventional cardiologists may have considered the accruing insights from randomised stent trials which, in the current guidelines, led to a strong recommendation of drug-eluting stents (DES) over bare metal stents for any PCI³. Still, in patients with a high bleeding risk or a strong preference for short DAPT, operators may favour new-generation DES with proven safety after a short period of DAPT or polymer-free DES.

During the last decade, there has been a shift from overnight stay to same-day discharge after PCI, if performed in an elective setting. A randomised controlled trial has found similar patientreported and clinical outcomes after same-day and next-day discharge²². In addition, 80% and 68% of the patients randomised to same-day and next-day discharge, respectively, stated that they would prefer same-day discharge after a potential future PCI. At 30-day follow-up, only 9% of patients in the same-day discharge group indicated a preference for overnight stay, if they had another PCI²². The results confirmed the findings of a previous observational study which reported that 89% of all patients who were treated with transradial PCI and discharged on the same day were satisfied with same-day discharge23. Hence, after uncomplicated elective PCI, the vast majority of patients prefer same-day discharge. Nevertheless, identifying the few patients who do not feel ready for same-day discharge and offering them an overnight stay is likely to increase the overall patient satisfaction, prevent unnecessary traumatisation, and facilitate potential future interventions in this particular patient group.

It has been advocated that patients should be more involved in research projects and outcomes, and that patient-reported outcome metrics should be defined and validated in the cardiovascular field^{1,24}. A study that aimed to identify outcomes that patients perceive as important following PCI identified feeling and function outcomes and the ability to perform usual activities as particularly important²⁵. This highlights that the patients' perspective on desirable outcomes of treatment may differ significantly from the physicians' perspective. Many clinical trials in the field of interventional cardiology assess a composite clinical endpoint, which means that reaching any of the constituent endpoints qualifies for reaching the composite endpoint. An example may be "target lesion failure", a composite endpoint of safety and efficacy, consisting of repeat revascularisation of an initially treated target lesion, experiencing a target vessel myocardial infarction, or death from cardiac (or unknown) causes. The PRECORE study assessed the perspective regarding such composite endpoints of patients who underwent PCI or CABG17. Patients attributed different weight to the individual constituent endpoints and considered the need for repeat PCI within one year as least severe, while a major stroke causing permanent disability was considered even worse than a fatal outcome. This suggests that many patients fear a loss of mobility and independence above death¹⁷. Other research also revealed that patients do not consider constituent endpoints of composite clinical endpoints equal²⁶. Hence, there is sufficient evidence that the current practice of most

clinical trials does not reflect patients' preference. Consequently, we feel that the time has come to stimulate and encourage a shift in thinking that may lead to the use of importance weight-adjusted composite endpoints in future clinical trials²⁷.

In conclusion, most patients appreciate their active involvement in the process of clinical decision making. Under appropriate circumstances, shared decision making can be applied when discussing various issues related to coronary angiography or PCI. For that purpose, patients need to be motivated and well informed. During the informed consent, there is a risk of misperception that can be reduced by the use of audiovisual tools and decision aids. However, this requires physicians to be willing to modify established practice patterns and embrace the true potential of decision aids. The duration of hospitalisation and recovery and the strain related to certain interventions are relevant to many patients. From the patients' perspective, these factors may affect preference for treatment even more than favourable results of clinical studies. It is desirable that physicians consider the treatment preference of the informed patient and find in true partnership with the patient the most suitable therapeutic approach. Finally, as most clinical trial results do not reflect the patients' perspective, there is a need for a shift towards trials with importance weight-adjusted composite clinical endpoints. Such trials will provide findings from the patients' perspective, and the study results will be truly relevant to them.

Thus, involving patients' preferences in the field of interventional cardiology has the potential to improve treatment significantly. This process has just begun and will not be easy, but is worth the endeavour.

Conflict of interest statement

C. von Birgelen reports institutional research grants to the Research Department of Thoraxcentrum Twente from Abbott Vascular, Biotronik, Boston Scientific, and Medtronic, unrelated to this manuscript. The other author has no conflicts of interest to declare.

References

1. Barbato E, Mauri J, Byrne RA, Roffi M, Fitzsimons D, Toth G, Capodanno D, Mehilli J, Petronio AS, Chieffo A, Vardas P, Naber CK, Dudek D, Haude M, Baumbach A. Patient focus in interventional cardiology: proceedings of the 2018 summit of the European Association of Percutaneous Cardiovascular Interventions (EAPCI). *EuroIntervention.* 2019;14:1720-3.

2. Knuuti J, Wijns W, Saraste A, Capodanno D, Barbato E, Funck-Brentano C, Prescott E, Storey RF, Deaton C, Cuisset T, Agewall S, Dickstein K, Edvardsen T, Escaned J, Gersh BJ, Svitil P, Gilard M, Hasdai D, Hatala R, Mahfoud F, Masip J, Muneretto C, Valgimigli M, Achenbach S, Bax JJ; ESC Scientific Document Group. 2019 ESC Guidelines for the diagnosis and management of chronic coronary syndromes. *Eur Heart J.* 2020;41:407-77.

3. Neumann FJ, Sousa-Uva M, Ahlsson A, Alfonso F, Banning AP, Benedetto U, Byrne RA, Collet JP, Falk V, Head SJ, Jüni P, Kastrati A, Koller A, Kristensen SD, Niebauer J, Richter DJ, Seferović PM, Sibbing D, Stefanini GG, Windecker S, Yadav R, Zembala MO. 2018 ESC/EACTS Guidelines on myocardial revascularization. *EuroIntervention*. 2019;14:1435-534.

4. Rothberg MB. PCI for stable angina: A missed opportunity for shared decision-making. *Cleve Clin J Med.* 2018;85:105-21.

5. Beach MC, Sugarman J. Realizing Shared Decision-making in Practice. *JAMA*. 2019 Jul 25. [Epub ahead of print].

6. Tak HJ, Ruhnke GW, Meltzer DO. Association of patient preferences for participation in decision making with length of stay and costs among hospitalized patients. *JAMA Intern Med.* 2013;173:1195-205.

7. Kok MM, Weernink MGM, von Birgelen C, Fens A, van der Heijden LC, van Til JA. Patient preference for radial versus femoral vascular access for elective coronary procedures: The PREVAS study. *Catheter Cardiovasc Interv.* 2018;91:17-24.

8. Goff SL, Mazor KM, Ting HH, Kleppel R, Rothberg MB. How cardiologists present the benefits of percutaneous coronary interventions to patients with stable angina: a qualitative analysis. *JAMA Intern Med.* 2014;174:1614-21.

9. Rothberg MB, Sivalingam SK, Kleppel R, Schweiger M, Hu B, Sepucha KR. Informed Decision Making for Percutaneous Coronary Intervention for Stable Coronary Disease. *JAMA Intern Med.* 2015;175:1199-206.

10. Rothberg MB, Sivalingam SK, Ashraf J, Visintainer P, Joelson J, Kleppel R, Vallurupalli N, Schweiger MJ. Patients' and cardiologists' perceptions of the benefits of percutaneous coronary intervention for stable coronary disease. *Ann Intern Med.* 2010;153:307-13.

11. Schwarzman L, Miron-Shatz T, Maki K, Hsueh L, Liu E, Tarashandegan D, Mendez F, Vidovich MI. Shared Decision-Making in Femoral Versus Radial Cardiac Catheterization. *Am J Cardiol.* 2019;124:190-4.

12. Dobler CC, Sanchez M, Gionfriddo MR, Alvarez-Villalobos NA, Singh Ospina N, Spencer-Bonilla G, Thorsteinsdottir B, Benkhadra R, Erwin PJ, West CP, Brito JP, Murad MH, Montori VM. Impact of decision aids used during clinical encounters on clinician outcomes and consultation length: a systematic review. *BMJ Qual Saf.* 2019;28:499-510.

13. Doll JA, Jones WS, Lokhnygina Y, Culpepper S, Parks RL, Calhoun C, Au DH, Patel MR. PREPARED study: A Study of Shared Decision-Making for Coronary Artery Disease. *Circ Cardiovasc Qual Outcomes*. 2019;12:e005244.

14. Coylewright M, Dick S, Zmolek B, Askelin J, Hawkins E, Branda M, Inselman JW, Zeballos-Palacios C, Shah ND, Hess EP, LeBlanc A, Montori VM, Ting HH. PCI Choice Decision Aid for Stable Coronary Artery Disease: A Randomized Trial. *Circ Cardiovasc Qual Outcomes*. 2016;9:767-76.

15. Coylewright M, O'Neill ES, Dick S, Grande SW. PCI Choice: Cardiovascular clinicians' perceptions of shared decision making in stable coronary artery disease. *Patient Educ Couns.* 2017;100:1136-43.

16. Kim C, Hong SJ, Ahn CM, Kim JS, Kim BK, Ko YG, Choi D, Jang Y, Hong MK. Patient-Centered Decision-Making of Revascularization Strategy for Left Main or Multivessel Coronary Artery Disease. *Am J Cardiol.* 2018;122:2005-13.

17. Vaanholt MCW, Kok MM, von Birgelen C, Weernink MGM, van Til JA. Are component endpoints equal? A preference study into the practice of composite endpoints in clinical trials. *Health Expect.* 2018;21:1046-55.

18. Ohlow MA, Farah A, Kuntze T, Lauer B. Patients' preferences for coronary bypass grafting or staged percutaneous coronary intervention in multi-vessel coronary artery disease. *Int J Clin Pract.* 2018;72:e13056.

19. Masdjedi K, Daemen J, Diletti R, Wilschut J, Utens E, de Jaegere PP, Lemmert ME, Kappetein AP, Zijlstra F, van Domburg R, Van Mieghem NM. A case-vignette based assessment of patient's perspective on coronary revascularization strategies, the OPINION study. *J Cardiol.* 2018;72:149-54.

20. Qintar M, Chhatriwalla AK, Arnold SV, Tang F, Buchanan DM, Shafiq A, Pokharel Y, deBronkart D, Ashraf JM, Spertus JA. Beyond restenosis: Patients' preference for drug eluting or bare metal stents. *Catheter Cardiovasc Interv.* 2017;90:357-63.

21. Chhatriwalla AK, Decker C, Gialde E, Catley D, Goggin K, Jaschke K, Jones P, deBronkart D, Sun T, Spertus JA. Developing and Testing a Personalized, Evidence-Based, Shared Decision-Making Tool for Stent Selection in Percutaneous Coronary Intervention Using a Pre-Post Study Design. *Circ Cardiovasc Qual Outcomes*. 2019;12:e005139.

22. Kim M, Muntner P, Sharma S, Choi JW, Stoler RC, Woodward M, Mann DM, Farkouh ME. Assessing patient-reported outcomes and preferences for same-day discharge after percutaneous coronary intervention: results from a pilot randomized, controlled trial. *Circ Cardiovasc Qual Outcomes*. 2013;6:186-92.

23. Ziakas A, Klinke P, Fretz E, Mildenberger R, Williams MB, Siega AD, Kinloch RD, Hilton JD. Same-day discharge is preferred by the majority of the patients undergoing radial PCI. *J Invasive Cardiol.* 2004;16:562-5.

24. Kok MM, von Birgelen C, Lam MK, Löwik MM, van Houwelingen KG, Stoel MG, Louwerenburg JH, de Man FH, Hartmann M, Doggen CJ, van Til JA, IJzerman MJ. Patient preference regarding assessment of clinical follow-up after percutaneous coronary intervention: the PAPAYA study. *EuroIntervention*. 2016;11:1487-94.

25. Ayton DR, Barker AL, Peeters GMEE, Berkovic DE, Lefkovits J, Brennan A, Evans S, Zalcberg J, Reid C, Stoelwinder JJ, McNeil J. Exploring patient-reported outcomes following percutaneous coronary intervention: A qualitative study. *Health Expect*. 2018;21:457-65.

26. Ahmad Y, Nijjer S, Cook CM, El-Harasis M, Graby J, Petraco R, Kotecha T, Baker CS, Malik IS, Bellamy MF, Sethi A, Mikhail GW, Al-Bustami M, Khan M, Kaprielian R, Foale RA, Mayet J, Davies JE, Francis DP, Sen S. A new method of applying randomised control study data to the individual patient: A novel quantitative patient-centred approach to interpreting composite end points. *Int J Cardiol.* 2015;195:216-24.

27. Abdallah MS, Wang K, Magnuson EA, Spertus JA, Farkouh ME, Fuster V, Cohen DJ; FREEDOM Trial Investigators. Quality of life after PCI vs CABG among patients with diabetes and multivessel coronary artery disease: a randomized clinical trial. *JAMA*. 2013;310:1581-90.